Author's response to reviews

Title: Realising the potential of the family history in risk assessment and primary prevention of coronary heart disease in primary care: ADDFAM study protocol [ISRCTN17943542]

Authors:

Nadeem Qureshi (nadeem.qureshi@nottingham.ac.uk)
Sarah Armstrong (sarah.armstrong@nottingham.ac.uk)
Paula Saukko (p.saukko@lboro.ac.uk)
Tracey Sach (t.sach@uea.ac.uk)
Jo Middlemass (jo.middlemass@nottspct.nhs.uk)
Phil Evans (philip.evans@pms.ac.uk)
Joe Kai (joe.kai@nottingham.ac.uk)

Version: 3 Date: 25 August 2009

Author's response to reviews:

Dear Sir/Madam,

Thank you for allowing time for the suggested revisions to be made to this manuscript.

In accordance with your instructions the following revisions have been made:

1. a statement regarding the ethical committee that gave approval has been included in the Methods section
2. the trial registration number has been removed from the title
3. a signed copy of the confirmation of funder letter has been attached (and also emailed to you)

I hope you find these revisions meet your requirements. If you require any further amendments please do not hesitate to contact me.

Yours faithfully

Dr Nadeem Qureshi